Author's response to reviews

Title: Legionella spp. and legionellosis in Southeastern Italy: disease epidemiology and environmental surveillance in community and health care facilities

Authors:

Christian Napoli (c.napoli@igiene.uniba.it)
Fabrizio Fasano (c.napoli@igiene.uniba.it)
Roberta Iatta (c.napoli@igiene.uniba.it)
Giovanna Barbuti (c.napoli@igiene.uniba.it)
Teresa Cuna (c.napoli@igiene.uniba.it)
Maria Teresa Montagna (c.napoli@igiene.uniba.it)

Version: 7 Date: 25 October 2010

Author's response to reviews:

Dear Editor,

we made all the Major revisions (it is required that the author makes these changes) that the Editorial Production Team required. With regard to minor revisions (the editorial team can make these changes for the author), we removed the line numbers, the highlighting and the vertical lines from tables, but we were not able to further minimise white space around the figures, we hope the editorial team could do this for us.

Thank you and best regards

The Authors